Author's response to reviews

Title: Patient experiences, attitudes and expectations towards receiving information about anti-TNF: A qualitative study - "It could give me two heads and I'd still try it!"

Authors:

Paul Arkell (paularkell@doctors.org.uk)
Sarah Ryan (Sarah.Ryan@uhns.nhs.uk)
Ann Brownfield (Ann.Brownfield@uhns.nhs.uk)
Anthony Cadwgan (Anthony.Cadwgan@uhns.nhs.uk)
Jon Packham (Jon.Packham@uhns.nhs.uk)

Version: 4 Date: 24 April 2013

Author's response to reviews:

Dear Editor,

The following amendments have been made:

1) Sentence in methods section has been changed from "Written consent was obtained from all participants according to the Declaration of Helsinki" to "Written consent was obtained for publication of individual clinical details from all participants, according to the Declaration of Helsinki".

2) Consent forms for all participants are provided as an appendix.

Yours Faithfully,

Dr Paul Arkell